Author’s response to reviews

Title: Traditional Chinese Medicine Use Is Associated with Lower End-Stage Renal Disease and Mortality Rates among Patients with Diabetic Nephropathy: A Population-Based Cohort Study

Authors:
Hsing-Yu Chen (b8705016@gmail.com)
Heng-Chih Pan (u8801006@gmail.com)
Yung-Chang Chen (cyc2356@gmail.com)
Yu-Chun Chen (yuchn.chen@googlemail.com)
Yi-Hsuan Lin (cynthiajobo@gmail.com)
Sien-Hung Yang (dryang@adm.cgmh.org.tw)
Jiun-Liang Chen (a12015@adm.cgmh.org.tw)
Hau-Tieng Wu (hauwu@math.duke.edu)

Version: 5 Date: 20 Mar 2019

Author’s response to reviews:
The manuscript is revised and re-written according to the comment.